Check for updates

# G OPEN ACCESS

**Citation:** Sanders AM, Dixon R, Stuck L, Kelly M, Woods G, Muheki EM, et al. (2021) Evaluation of facial cleanliness and environmental improvement activities: Lessons learned from Malawi, Tanzania, and Uganda. PLoS Negl Trop Dis 15(11): e0009962. https://doi.org/10.1371/journal. pntd.0009962

**Editor:** Jeremiah M. Ngondi, RTI International, UNITED REPUBLIC OF TANZANIA

Received: April 24, 2021

Accepted: November 1, 2021

Published: November 29, 2021

**Copyright:** © 2021 Sanders et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** The author(s) received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

RESEARCH ARTICLE

# Evaluation of facial cleanliness and environmental improvement activities: Lessons learned from Malawi, Tanzania, and Uganda

Angelia M. Sanders<sup>1</sup>\*, Ruth Dixon<sup>2</sup>, Logan Stuck<sup>3</sup>, Michaela Kelly<sup>2</sup>, Geordie Woods<sup>2</sup>, Edridah M. Muheki<sup>4</sup>, Gilbert Baayenda<sup>4</sup>, Michael Masika<sup>5</sup>, Holystone Kafanikhale<sup>5</sup>, Upendo Mwingira<sup>6</sup>, Leah Wohlgemuth<sup>2</sup>

The Carter Center, Atlanta, Georgia, United States of America, 2 Sightsavers, London, United Kingdom,
 Amsterdam Institute for Global Health and Development, Amsterdam, Netherlands, 4 Uganda Ministry of Health, Kampala, Uganda, 5 Malawi Ministry of Health, Lilongwe, Malawi, 6 Tanzania Ministry of Health, Dar es Salaam, Tanzania

\* Angelia.sanders@cartercenter.org

# Abstract

The World Health Organization promotes the SAFE (Surgery, Antibiotics, Facial cleanliness, and Environmental improvements) strategy for trachoma control and prevention. The F&E components of the strategy focus on promotion of healthy hygiene and sanitation behaviors. In order to monitor F&E activities implemented across villages and schools in Malawi, Tanzania, and Uganda, an F&E Monitoring and Evaluation (FEME) framework was developed to track quarterly program outputs and to provide the basis for a pre and post evaluation of the activities. Results showed an increase in knowledge at the school and household levels, and in some cases, an increase in presence of hand/face washing stations. However, this did not always result in a change in trachoma prevention behaviors such as facial cleanliness or keeping compounds free of human feces. The results highlight that the F&E programs were effective in increasing awareness of trachoma prevention but not able to translate that knowledge into changes in behavior during the time between pre and post-surveys. This study also indicates the potential to improve the data collection and survey design and notes that the period of intervention was not long enough to measure significant changes.

# Author summary

Trachoma control programs promote facial cleanliness, use of latrines, and emphasizes the importance of access to water as means to reduce trachoma transmission. To address these areas, various country ministries and supporting non-governmental organizations (NGOs) support the implementation of tailored behavior change programs. The process and results of evaluating these types of programs in Malawi, Tanzania, and Uganda are described here. The goal of this manuscript is to provide trachoma program managers and supporting NGOs with insights, recommendations, and data collection tools that could be used to support their efforts to conduct monitoring and evaluation of their current or future trachoma prevention activities.

#### Introduction

#### Trachoma and water, sanitation, and hygiene

Trachoma, the result of ocular infection with the bacterium *Chlamydia trachomatis*, is the leading infectious cause of blindness worldwide [1]. Over time, repeated infection can cause the inside of the eyelid to become scarred and the eyelid and eyelashes to turn inward. Left untreated, the painful condition of the in-turned eyelashes scrapping the cornea can cause the individual to become irreversibly visually impaired or blind. Children have been documented to have higher levels of trachoma infection compared to other age groups, with the prevalence of active trachoma greatest among preschool age children [2–4]. The progression of the disease to the advanced blinding stage occurs over time with the onset of visual impairment typically occurring in those aged 35 years and over [5].

Transmission of the bacterium Chlamydia trachomatis can occur when fingers and fomites such as clothes and towels come into contact with infected ocular and nasal secretions from one individual and then contact the eyes of another individual [6]. Flies have also been documented to transmit Chlamydia trachomatis when the fly feeds on the ocular and nasal secretions of an infected person and then lands on the eyes of another person [7]. These same flies are strongly attracted to odors produced by human feces and lay their eggs on exposed feces on the ground [8]. Trachoma frequently affects poor and marginalized populations that typically sleep in close quarters, experience overcrowding, and lack adequate access to water and sanitation [1,9,10]. It has been challenging to determine the relative importance of one route of transmission compared to others, and it is unlikely that any single route of transmission is responsible for all trachoma transmission [6,11]. Despite not knowing the extent that Water, Sanitation, and Hygiene (WASH) programs impact trachoma prevalence within communities, certain human behaviors have been demonstrated to reduce the risk of trachoma transmission. First, reducing exposed human feces, through methods such as using pit latrines, reduces fly breeding grounds [7,8]. Second, avoiding sharing cloth with infected ocular or nasal discharge reduces the transmission of the bacterium between people [11]. Third, keeping one's face clean from ocular and nasal secretions reduces the attractiveness of the face to the flies that feed on the secretions for nutritional purposes while simultaneously transmitting the bacterium [7]. Lastly, increasing communities' sufficient and reliable access to water helps increase the likelihood that residents will be able to keep their faces, clothes, and bedding clean thereby reducing risk [12]. It is because of the transmission dynamics and progression of the disease that the World Health Organization (WHO) promotes the SAFE strategy (Surgery, Antibiotics, Facial cleanliness, and Environmental improvements) for trachoma control and prevention [1]. The F&E components of the strategy are focused on promoting healthy hygiene and sanitation behaviors and have a significant overlap with WASH activities. Given that the F&E components are outcomes, there are diverse types of interventions that can be used to achieve these outcomes. The lack of standardization and consistency in how these interventions are designed and implemented present a challenge in comparing across settings and clearly demonstrating effectiveness.

#### Selection of F&E interventions

The trachoma control programs in Malawi, Tanzania, and Uganda received funding from The Queen Elizabeth Diamond Jubilee Trust (hereafter referred to as the Trust) for F&E activities, with additional funding provided to Tanzania from the United Kingdom's Department For International Development (DFID) [13,14]. In order to guide decision making about how these funds should be spent the Ministry of Health in each country used a multi-step process. First, a situational analysis of WASH and trachoma programming within each country was conducted in order to understand the partners, resources, and existing WASH, trachoma, and F&E activities already taking place at the regional and district levels [15]. Second, the "All You Need for F&E" International Coalition for Trachoma Control (ICTC) toolkit was used to guide an F&E stakeholder workshop [16]. Representatives from ministries of water and sanitation, health, and education participated in this workshop alongside representatives from NGOs from the WASH, neglected tropical diseases (NTDs), and research sectors. Workshop participants identified a list of F&E activities and then participants narrowed down the number of activities based on available donor funding and what they felt should be prioritized and could be achievable in three years. Table 1 shows an overview of the different activities chosen for implementation by region within each country. A more detailed description of activities

Country	Region	Activities selected
Malawi	Central	CLTS+; SLTS+; Hand and face washing stations; and BCC at community and school level: Radio, drama, songs, and print, work with school health clubs and district level committees.
	Southern	CLTS+, SLTS+, Hand and face washing stations, BCC at community and school level: Radio, drama, songs, and print, work with school health clubs and district level committees
Tanzania	Arusha and Dodoma	School WASH through School Health Clubs; SLTS+; Social marketing; and Trachoma focused WASH education in the school curricula and mandates of teachers.
	Lindi, Manyara, Pwani	CLTS+; Social marketing to promote face washing stations; "Daily nudges" towards the adoption of healthy hygiene and sanitation practices; and Supporting National Sanitation Campaign of the Ministry of Health and Social Welfare.
Uganda	Karamoja	CLTS+; Hand and face washing stations; BCC through Mother Care Groups and school health clubs; and Adaption of school curricula to include trachoma messaging.
	Busoga	Access to water through new safe water projects (funded by alternative source), strengthening water committees; BCC: training WASH promoters at parish level to work in communities, schools, community theatre, and radio; and Adaption of school curricula to include trachoma messaging.

Table 1. Country selected F&E activities by region.

CLTS+ = community led total sanitation with trachoma messaging; SLTS+ = school led total sanitation with trachoma messaging; BCC = behavior change communication. Note: Community Led Total Sanitation (CLTS) is a recognized methodology for mobilizing communities to completely eliminate open defecation. Communities are facilitated to conduct their own appraisal and analysis of open defecation and define their own actions including their own building of toilets and hand and face washing stations by the communities themselves to make the community Open Defecation Free (ODF) and achieve the 'ODF-status'. For this project, we adapted the CLTS-methodology to a school setting. School Led Sanitation (SLTS) methods encourage pupils to analyze their home situation related to WASH. The WASH situation at home is discussed with the target audience, visits made and jointly solutions are sought to improve their home situation, as part of a real-life learning cycle. Through this approach we increase the outreach of the behavior change communication from the schools into the community. As part of these projects, we made sure that the CLTS and SLTS approaches included an element on trachoma, a strong emphasis on sanitation, and a focus on hand and face washing.

implemented within each country is provided in **S1 Text**. Ministries of health, water, and education, and supporting implementing partners implemented these interventions from July 2015 to March 2018 in Malawi, May 2016 to March 2018 in Tanzania, and May 2015 to March 2018 in Uganda. While F&E interventions targeted districts with a trachomatous inflammation-follicular (TF) prevalence over 5% (in children ages one to nine years), not all selected program activities were conducted in all villages within the targeted districts due to lack of funding and implementing partner availability.

#### Monitoring and evaluating F&E

An F&E Monitoring and Evaluation (FEME) framework was developed for each country to assist with conducting quarterly monitoring of F&E activities and to provide the basis for a pre and post evaluation of F&E activities. The indicators used within each country's FEME included a combination of WASH and NTD indicators identified during a Delphi consultative process [17] and country specific indicators requested by the Ministry of Health. Country specific FEMEs are provided in supplemental information (<u>S1, S2</u> and <u>S3</u> Tables).

The FEME can be divided into two parts: 1) the logical framework that includes outcome and output indicators; and 2) program implementation activities. Theoretically implementation of activities leads to the achievement of the desired outputs and outcomes reflected by changes in their indicators. For example, conducting community meetings about trachoma (activity) should result in an increase in the percentage of people who have knowledge of hygiene practices in relation to trachoma prevention (output) which thereby contributes to an increase in the percentage of children with clean faces (outcome). Throughout the life of the Trust funded project within the three countries the F&E activities were reported on a quarterly basis. For measuring progress on achieving outcomes and outputs, each country conducted a pre and post-survey.

# Objectives of the paper

The purpose of this paper is threefold: 1) present the methods and results of the pre and post surveys in each of the three countries; 2) discuss challenges and successes of the survey process and indicator measures selected; and 3) provide recommendations based on this experience for implementing F&E monitoring and evaluation mechanisms.

# Methods

# **Ethics statement**

Ethical approval was obtained for the pre and post-surveys in all three countries. In Malawi from the National Health Sciences Research Committee (Ref: #16/12/1700), in Tanzania from the National Institute for Medical Research (NIMR) National Health Ethics Review Committee (NatHREC) (Ref: NIMR/HQ/R.8a/Vol. IX/2405), and in Uganda from the Uganda National Council for Science and Technology (Ref: HS 2166). The study conducted in Uganda had additional ethical approval from Emory University (eIRB#: IRB00093647). All survey participants gave written informed consent prior to participating, head teachers provided written informed consent on behalf of school children who individually assented to take part. Assent was documented.

# **Evaluation units**

F&E programmatic intervention regions were used as evaluation units (EUs) with sampling of villages from a sample frame of all intervention villages. Where the scale of interventions was

			<b>Community Evaluation Unit</b>	ts		
Country	Regions within an EU	Number intervention villages	Sample target Comm/HH	Pre-survey 2017	Post-survey 2018	Intervention period evaluated
Malawi	Central	390	36	January	June	16 months
Malawi	Southern	258	36			
Tanzania	Lindi, Manyara, Pwani	68	60	March	May	13 months
Uganda	Karamoja	1592	42	February	November	20 months
Uganda	Busoga	4702	40			
			School Evaluation Units			
Country	Regions within an EU	Number intervention schools	Sample target schools/ SAC	Pre-survey 2017	Post-survey 2018	Intervention period evaluated
Malawi	Central	107	27	January	June	16 months
Malawi	Southern	246	28			
Tanzania	Arusha and Dodoma	20	22	March	May	13 months
Uganda	Karamoja	24	15	February	November	20 months

#### Table 2. Evaluation Units (EUs).

Comm = communities; HH = households; SAC = school age children

https://doi.org/10.1371/journal.pntd.0009962.t002

small (Tanzania) multiple regions were grouped into a single EU. For regions where there was both school and community programming the EUs were considered separately (Table 2).

In all three countries, implementation of F&E activities began prior to the pre-surveys being conducted. Pre-surveys were conducted in 2017 and post-surveys conducted in 2018. Fig 1 provides a visual of the regions within each country where community and school activities took place and the location of the villages and schools that were selected for evaluation.

#### Sampling

The sample size calculations were performed in STATA using the "sampsi" function. Sample size was based on the indicator 'percentage of children with a clean face' assuming 50% at baseline and the desire to detect a policy relevant change of 10 percentage points. We used 80% power in all calculations, incorporated a design effect of 2 (villages) and 1.5 (schools) and an anticipated non-response rate of 15%.

# **Community sampling**

Household data was collected using a population-based survey following a two-stage cluster sampling methodology. In each community EU all villages targeted for F&E interventions were listed along with their estimated population sizes. The village was the primary sampling unit and was selected using probability proportional to size. The same villages were surveyed at both pre and post-survey. The secondary sampling unit, the household, was randomly selected using a household listing approach. Households that did not have at least one adult ( $\geq$ 18 years in Tanzania and Uganda,  $\geq$ 15 years in Malawi) who identified as the primary care giver and at least one child under nine years old were excluded from participating and replaced. Every effort was made to collect data from the selected households, which were visited three times before being replaced in cases of recurrent non-occupancy. Non-participating selected households were also replaced. Twelve villages in Lindi which had both school and community programming were excluded from the sample frames for this EU.

#### School sampling

In Uganda schools were randomly selected from a list of all intervention schools in the EU. In Malawi they were selected using probability proportionate to size. In Tanzania, all intervention schools were included in the survey. In all three countries, in each selected school, two classes were randomly selected from all classes targeted with the trachoma F&E interventions. Within each of these two classes, 21 students were randomly selected for questionnaires. The same schools were used in the pre and post-surveys.

#### Data collection

The surveys were designed and conducted through collaboration of partners in each country. Independent teams were recruited for the data collection who were blind to the interventions conducted and other indicators that would be generated from the data. Household questionnaires were conducted with the primary caregiver. All present household residents were directly observed for signs of ocular or nasal discharge [18]. As a proxy indicator for hand and face washing, data was collected on the presence and functionality of hand/face washing stations [19]. Direct observation of the availability of handwashing and toilet facilities was conducted. School questionnaires were conducted with head teachers along with observations of availability of toilet and handwashing facilities. Individual questionnaires were conducted with students along with observations of presence of ocular or nasal discharge on the face. Student hand and face washing behaviors were observed for a period of at least three hours in each school. In order to reduce bias we ensured that all observers/research assistants were from the respective countries, spoke the local language, and observed the events as discretely as possible. We also did not tell the schools exactly what we were observing (i.e. we did not say we were there to observe hand and face washing practices specifically). The extended observation time (3 hours) was selected in an effort to capture different opportunities in which children would wash their hands and face and also allowed the children to get used to the observer being present. There is always a risk of observer bias in the study, and we recognized this as a limitation and made as many efforts as possible to limit these biases. Household questionnaire guides were translated into the predominant local language. All questionnaires were pilot tested in the language in which it was to be conducted and revisions were made to increase clarity. School questionnaires with the head teacher were conducted in English while student ones were conducted in the local language. The English version of the questionnaires and observational data collection tools are provided in supporting information (S4, S5 and S6 Tables). Table 3 documents the different types of data collection methods used in schools and communities. With the exception of the student 'hand/face washing behavior observation', all observations were embedded components of the school or household questionnaire.

#### Data management and analysis

Data was collected electronically using a purpose-built Open Data Kit-based Android smartphone application- LINKS for pre-survey and CommCare for post-survey [20,21]. Data was downloaded, imported to STATA13 and weighted according to sampling design and analyzed for a difference between pre and post-surveys within EUs using chi squared statistical test. A p-value of <0.05 was used to attribute a statistical difference between pre and post.

# Results of pre and post surveys

The implementation of F&E activities and quantitative evaluations of those activities were not intended as a one size fits all approach; however, common themes emerged across the three



Fig 1. Location of F&E interventions and sampled villages and schools for pre and post-survey. Base map is from NaturalEarth for administrative boundary <a href="https://www.naturalearthdata.com/downloads/10m-cultural-vectors/">https://www.naturalearthdata.com/downloads/10m-cultural-vectors/</a> and for lake layer <a href="https://www.naturalearthdata.com/downloads/10m-physical-vectors/">https://www.naturalearthdata.com/downloads/10m-cultural-vectors/</a> and for lake layer <a href="https://www.naturalearthdata.com/downloads/10m-physical-vectors/">https://www.naturalearthdata.com/downloads/10m-cultural-vectors/</a> and for lake layer <a href="https://www.naturalearthdata.com/downloads/10m-physical-vectors/">https://www.naturalearthdata.com/downloads/10m-physical-vectors/</a>.

https://doi.org/10.1371/journal.pntd.0009962.g001

countries. In order to clearly present results of the pre and post-surveys, results are organized into the thematic groupings of WASH infrastructure, trachoma knowledge, and F&E related behaviors and are presented first for school EUs and then community EUs.

#### **School EU Results**

Key results from each school EU within each country are provided in Table 4.

Only the EU in Tanzania showed a significant increase in the percentage of schools that had hand washing facilities with soap with 4.8% (Confidence Interval (CI): 0.6-30.7) at baseline and 35.0% (CI: 17.3–58.0) at post-survey (P = 0.043). In Uganda, there was significant increase in the percentage of schools that had at least one clean latrine for both boys and girls, with 27.8% (CI: 9.9–57.3) at baseline and 70.8% (CI: 39.1–90.2) at post-survey (P = 0.039).

There was minimal positive behavior change across the three countries. Only the two school EUs in Malawi showed a significant change in the percentage of school compounds free of human feces, with those in the Central region increasing from 19.6% (CI: 6.0-48.1) at baseline to 92.6% (72.6–98.3) at post-survey (P < 0.01) and those in the Southern region increasing

Evaluation Unit Type	Data Collection Method	Sample Characteristic	Pre/ Post Survey	Sample Size by EU									
			Ma	alawi	Tanza	nia	Uganda						
				Central	Southern	Lindi-Manyara- Pwani	Arusha- Dodoma	Karamoja	Busoga				
School	School questionnaire and	Schools	Pre	28	27		21	15					
	facility observation		Post	27	29		20	14					
School	Student questionnaire and	Schools/ students	Pre	28/639	27/649		21/676	15/630					
	face observation		Post	27/660	29/700		20/640	14/604					
School	Hand or face washing	Schools/	Pre	28/570	27/880		21/553	15/195					
	behavior observation	observed events	Post	27/1090	29/1623		20/1909	14/604					
Community	Household questionnaire and	Communities/	Pre	36/967	35/945	58/1692		43/1008	40/958				
	observation	households	Post	35/943	36/978	59/1696		43/1021	40/965				
Community	Household face observation	Households/ faces observed	Pre	967/ 4493	945/4509	1692/6818		1008/5206	958/ 5372				
			Post	943/ 2833	978/2817	1696/4914		1021/3052	965/ 3373				

#### Table 3. Achieved sample size by sample characteristic at pre and post-survey.

https://doi.org/10.1371/journal.pntd.0009962.t003

from 7.5% (CI: 1.9–25.0) at baseline to 74.2% (CI: 28.6–95.4) at post-survey (P < 0.01). However, 100% of surveyed schools in Tanzania were free from human feces at baseline which was maintained through the post-survey. In Karamoja, Uganda, there was a decrease in the

#### Table 4. Survey results by school evaluation unit at pre and post-survey.

			Mala	awi				Fanzania			Uganda	
	Central			Southern			Arusha and	Dodoma		Karamoja		
Schools	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р
WASH infrastructure												
% of schools with hand/face washing facilities with soap (1)	8.8 (2.7– 25.4)	28.4 (13.8– 49.6)	0.067	18.9 (4.2– 55.5)	9.2 (3.0– 25.0)	0.410	4.8 (0.6– 30.7)	35.0 (17.3– 58.0)	0.043	13.9 (2.9– 46.8)	15.3 (3.9– 44.2)	0.916
% of schools that have at least one clean latrine for both boys and girls (1)	44.8 (20.7– 71.7)	70.3 (46.0– 86.8)	0.157	61.0 (28.7– 85.8)	78.0 (46.0– 93.7)	0.395	Data not available			27.8 (9.9– 57.3)	70.8 (39.1– 90.2)	0.039
Behavior												
% of children with clean faces (no ocular or nasal discharge) among all children at school (2)	96.8 (93.2– 98.5)	95.1 (91.5– 97.2)	0.352	96.9 (92.6– 98.7)	93.6 (87.5– 96.9)	0.189	96.2 (88.8– 98.8)	94.7 (90.6– 97.1)	0.594	90.8 (81.2– 95.7)	87.6 (82.2– 91.5)	0.442
% of school children washing their faces when washing their hands during school day (3)	9.3 (4.8– 17.4)	6.6 (2.5– 16.5)	0.259	10.5 (4.2– 23.6)	5.1 (1.6– 14.8)	0.135	6.2 (2.3– 15.9)	9.1 (5.1– 15.9)	0.9	22.0 (12.2– 36.2)	13.0 (6.0– 25.9)	0.048
% of school compounds free of human feces (1)	19.6 (6.0– 48.1)	92.6 (72.6– 98.3)	<0.001	7.5 (1.9– 25.0)	74.2 (28.6– 95.4)	0.004	100.0 (100.0- 100.0)	100.0 (100.0- 100.0)	1.0	69.4 (37.1– 89.7)	90.3 (48.1– 98.9)	0.268
Trachoma knowledge												
% of school children who know at least one measure to prevent trachoma (2)	51.3 (44.0– 58.5)	81.7 (73.5– 87.8)	<0.001	62.8 (43.9– 78.5)	66.7 (60.6– 72.2)	0.67	30.2 (21.5– 40.6)	49.8 (41.8– 57.7)	0.004	49.9 (39.6- 60.1)	76.3 (61.9– 86.5)	0.004

Data collection method used: (1) school questionnaire and facility observation; (2) student questionnaire and face observation; (3) student hand/face washing behavior observation; CI = confidence interval; P = p-value

percentage of school children washing their faces when washing their hands during the school day with a baseline of 22.0% (CI: 12.2–36.2) decreasing to 13.0% (CI: 6.0–25.9) at post-survey (P = 0.048). There was no significant improvement in the percentage of students with a clean face between pre and post-survey; however, the baseline percentage of children with clean face was already high across the three countries ranging from 90.8% (CI: 81.2–95.7) in Karamoja, Uganda, to 96.9% (CI: 92.6–98.7) in the Southern EU of Malawi. All but the Southern region of Malawi had a significant increase in the percentage of school children who knew at least one measure to prevent trachoma.

#### **Community EU Results**

WASH Infrastructure. Key WASH infrastructure results from each EU within each country are provided in Table 5. Access to WASH infrastructure varied throughout the EUs. Only the Karamoja region of Uganda had a significant increase in the percentage of households found to have a hand/face washing station from 1.4% (CI: 0.7-2.8) at baseline to 10.1% (CI: 7.2–14.1) at post-survey (P < 0.001). In Tanzania, the percentage decreased from 13.4% (CI: 9.5–18.5) at baseline to 3.8% (CI: 2.0–7.1) at post-survey (P > 0.001). Though the Central region of Malawi did not increase the percentage of households with a hand/face washing station, they did increase the presence of soap at these washing stations. Presence of soap at the stations also increased in Karamoja and Busoga regions of Uganda, with an increase from 0.3% (CI: 0.1–0.9) at baseline to 2.3% (CI: 1.4–3.9) at post-survey (P = 0.001) in Karamoja and 2.1% (CI: 1.4-3.2) at baseline to 4.8% (CI: 2.3-9.5) at post-survey (P = 0.049) in Busoga. Only Karamoja, Uganda, had a significant change in the percentage of households with hand/face washing facilities that had water. The percentage of households with access to latrines increased in the Southern region of Malawi from 80.5% (CI: 73.3-86.1) to 88.7% (CI: 85.0-91.5) (P = 0.015) and Busoga, Uganda, from 92.3% (87.7-95.3) to 97.6% (CI: 95.6-98.7) (P > 0.003). The most dramatic increase in access to latrines was in the Karamoja region of Uganda with access increasing from 29.6% (CI: 20.2-41.0) at baseline to 93.6% (CI: 89.4-96.2) at post-survey (P < 0.001).

			Ma	lawi			r	Fanzania			Uganda					
	Central			Southern	ı		Lindi, M	anyara, Pw	vani	Karamoj	a		Busoga			
WASH infrastructure	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р										
% of households found to have a hand/face washing station (1)	19.6 (13.0– 28.6)	25.7 (18.6– 34.3)	0.277	25.7 (15.4– 39.6)	41.7 (29.6– 54.8)	0.08	13.4 (9.5– 18.5)	3.8 (2.0– 7.1)	0.001	1.4 (0.7– 2.8)	10.1 (7.2– 14.1)	<0.001	11.5 (7.8– 16.8)	11.8 (8.5– 16.3)	0.916	
% of households with hand/face washing facilities <i>with soap</i> (1)	3.7 (2.2– 6.1)	10.9 (7.2– 16.1)	0.001	7.5 (3.7– 14.5)	16.3 (9.7– 26.0)	0.244	2.1 (1.0- 4.4)	2.1 (1.2- 3.4)	0.987	0.3 (0.1- 0.9)	2.3 (1.4– 3.9)	0.001	2.1 (1.4– 3.2)	4.8 (2.3- 9.5)	0.049	
% of households with hand/face washing facilities <i>with water</i> (1)	13.1 (8.2– 20.2)	19.2 (13.4– 26.8)	0.174	21.6 (12.2– 35.3)	31.9 (20.3- 46.3)	0.358	3.0 (1.8– 4.9)	3.1 (1.6– 5.9)	0.9	0.5 (0.2– 1.2)	4.0 (2.5- 6.4)	<0.001	5.3 (3.7– 7.5)	8.4 (5.5– 12.8)	0.094	
% of households with access to a latrine (1)	81.4 (74.7– 86.7)	81.5 (76.1– 85.8)	0.986	80.5 (73.3– 86.1)	88.7 (85.0– 91.5)	0.015	78.4 (60.0– 89.8)	83.4 (64.1– 93.4)	0.625	29.6 (20.2– 41.0)	93.6 (89.4– 96.2)	<0.001	92.3 (87.7– 95.3)	97.6 (95.6– 98.7)	0.003	

 Table 5. WASH infrastructure survey results by community evaluation unit at pre and post-surveys.

Data collection method used: (1) Household questionnaire and observation

Trachoma knowledge. Key results on trachoma knowledge from EUs within each country is provided in Table 6. For purposes of this manuscript, three indicators were used to measure a change in trachoma knowledge. These include: percentage of household respondents who knew one or more symptoms of trachoma; percentage of household respondents who had seen or heard any message about trachoma; and percentage of household respondents who knew one or more ways on how trachoma spreads. The Southern region of Malawi had no significant change in these three indicators. In Central Malawi, there was only an increase in percentage of respondents who knew one or more symptoms of trachoma with an increase from 39.9% (CI: 33.5–46.7) at baseline to 51.5% (CI: 42.9–60.1) at post-survey (P = 0.035). In Tanzania, there was a significant increase in percentage of household respondents who knew symptoms of trachoma and how the trachoma disease spreads. The Karamoja and Busoga regions of Uganda had a significant increase in all three indicators (P < 0.001). Additionally, it was only the Busoga region of Uganda that had a significant change in perception of personal and family risk of trachoma, which increased from 26.1% (CI: 21.3-31.6) at baseline to 38.9% (CI: 34.3-43.7) at post-survey (P > 0.001). All other evaluation units stayed approximately the same percentage between pre and post-surveys.

**F&E related behavior.** Key results from EUs within each country are provided in Table 7 regarding trachoma behavior related indicators. Two indicators are reported here: clean face, defined as a face free from ocular and nasal discharge; and households free of human feces. The facial cleanliness indicator within households was broken down into three age groups: children nine years and younger; children 14 years and younger; and adults 15 years and above. This classification accommodates the nine years and younger that is typically used within trachoma programs to measure trachoma prevalence in children and the classification of adults as those 15 years and above for purposes of determining the advanced stage of trachoma (trachomatous trichiasis) in the adult population. Results across the facial cleanliness

			Ma	lawi				Tanzania		Uganda					
	Central			Souther	n		Lindi, N	lanyara, P	wani	Karamo	ja		Busoga		
Trachoma knowledge	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р									
% of households respondents who know one or more symptoms of trachoma (1)	39.9 (33.5– 46.7)	51.5 (42.9– 60.1)	0.035	42.5 (36.3– 49.1)	46.7 (40.4– 53.0)	0.999	34.7 (27.3– 42.9)	50.0 (44.5– 55.6)	0.003	50.2 (44.2– 56.3)	68.2 (62.7– 73.3)	<0.001	76.9 (69.7– 82.9)	97.3 (94.7– 98.6)	<0.001
% of households respondents who have seen or heard any message about trachoma (1)	8.2 (4.5– 14.5)	5.9 (4.6– 7.6)	0.303	5.5 (2.9– 10.2)	5.5 (2.7– 11.1)	0.262	3.7 (1.8– 7.6)	4.5 (2.4– 8.5)	0.676	7.7 (5.5– 10.6)	17.4 (13.4– 22.4)	<0.001	6.3 (4.4– 9.0)	13.7 (11.4– 16.3)	<0.001
% of households respondents who know one or more way on how trachoma disease spreads (1)	31.8 (23.6– 41.2)	39.8 (32.0– 48.1)	0.184	26.0 (19.7– 33.4)	32.2 (26.2– 38.8)	0.187	22.8 (18.0– 28.5)	35.0 (30.6– 39.7)	0.001	25.7 (21.6– 30.3)	56.1 (51.1– 60.9)	<0.001	44.6 (38.9– 50.4)	65.6 (59.9– 70.8)	<0.001
% of respondents thinking they and family are at risk of trachoma (1)	24.7 (19.2– 31.2)	25.9 (19.3– 33.9)	0.798	27.7 (21.6– 34.8)	23.3 (17.7- 30.0)	0.323	5.3 (4.0– 7.1)	6.9 (4.4– 10.5)	0.321	23.0 (18.3– 28.5)	24.8 (20.3– 29.9)	0.615	26.1 (21.3– 31.6)	38.9 (34.3– 43.7)	0.001

Table 6. Trachoma knowledge survey results l	y community evaluation unit a	t pre and post-surveys.
--	-------------------------------	-------------------------

Data collection method used: (1) Household questionnaire

			Ma	lawi			1	Fanzania		Uganda					
	Central			Souther	n		Lindi, M	lanyara, P	wani	Karamo	ja		Busoga		
Behavior	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р	Pre [% (95% CI)]	Post [% (95% CI)]	Р
% of children ages $\leq$ 9 years with clean faces (no ocular or nasal discharge) among all children in or near their home (2)	76.7 (71.6– 81.2)	63.9 (57.3– 70.0)	<0.001	76.3 (71.9– 80.1)	58.1 (52.5– 63.6)	<0.001	63.4 (54.9– 71.1)	62.3 (54.6– 69.4)	0.02	60.8 (55.6– 65.8)	48.7 (44.4– 52.9)	<0.001	62.4 (57.8– 66.8)	60.6 (57.0– 64.0)	0.002
% of children ages < 15 years with clean faces (no ocular or nasal discharge) among all children in or near their home (2)	75.2 (70.3– 79.5)	56.3 (50.3– 62.2)	<0.001	73.9 (69.1– 78.3)	52.0 (47.3– 56.6)	<0.001	60.6 (53.6– 67.1)	56.0 (49.6– 62.2)	0.02	60.2 (54.8– 65.3)	46.0 (42.1– 49.8)	<0.001	58.4 (53.4– 63.2)	52.0 (48.5– 55.4)	0.002
% of adults ages $\geq 15$ years with clean faces (no ocular or nasal discharge) (2)	69.9 (64.9– 74.4)	56.4 (52.8– 60.0)	<0.001	63.0 (54.2– 71.0)	61.2 (57.2– 65.1)	<0.001	67.1 (64.1– 70.0)	57.8 (55.0– 60.6)	0.02	72.0 (67.3– 76.2)	56.3 (52.2– 60.2)	<0.001	66.1 (62.2– 69.7)	55.6 (52.7– 58.4)	0.002
% of households free of human feces (1)	95.2 (92.5– 97.0)	95.6 (93.5– 97.0)	0.785	93.7 (89.6– 96.3)	95.6 (93.7– 96.9)	0.262	90.2 (81.0– 95.2)	87.6 (77.7– 93.5)	0.603	88.6 (84.0– 92.1)	86.8 (83.1– 89.7)	0.478	96.8 (94.6– 98.1)	92.8 (90.2- 94.8)	0.010

Data collection method used: (1) Household questionnaire and observation; (2) household face observation

https://doi.org/10.1371/journal.pntd.0009962.t007

indicators for all age groups showed a significant decrease in facial cleanliness. There was no significant change in percentage of households free of human feces, with the exception of Busoga, Uganda, where there was a decrease from 96.8% (CI: 94.6–98.1) at baseline to 92.8% (CI: 90.2–94.8) at post-survey. Despite little change in this indicator across the evaluation units, it is worth noting that the percentage of households free of human feces was above 85% at baseline for most regions, ranging from 88.6% (CI: 84.0–92.1) in Karamoja, Uganda, to 96.8% (CI: 94.6–98.1) in Busoga, Uganda. In Malawi, both regions were at 95.6% at post-survey.

#### Discussion

In order to meet the stated objectives of the paper, the discussion is broken down into three sub-sections. First, a discussion of country specific programmatic achievements in WASH infrastructure, trachoma knowledge, and F&E related behavior. Second, an examination of the challenges and successes in the survey design and indicator measures used, and finally recommendations for future implementers.

#### **Programmatic achievements**

**WASH infrastructure.** Data was collected on a range of WASH related indicators. For purposes of this paper, results and discussion focus on indicators that highlighted a house-hold's hygiene and sanitation related behaviors such as hand and face washing and latrine use, as these are believed to decrease the likelihood of trachoma transmission. As a proxy indicator for hand and face washing, data was collected on the presence and functionality of hand/face washing stations. The presence of soap and water were assumed to show an increased likelihood that the household was using the hand/face washing station. It was only in the Karamoja

region of Uganda that there was a significant increase in households with hand/face washing stations and the presence of soap and water at those locations. Despite the statistically significant increases in Karamoja and a few of the other regions for select WASH indicators, programmatically the results are not encouraging. At post-survey, the overall percentages of households with hand/face washing stations ranged from a low of 3.8% in Tanzania to a high of 41.7% in the Southern region of Malawi. Hand/face washing stations with water, a likely sign of their proper use, ranged from a low of 4.0% in Karamoja to a high of 31.9% in the Southern region of Malawi. These results highlight that even where there were hand/face washing stations, the percentage that had soap and/ or water was much smaller. This could signify that simple presence of hand/face washing stations did not guarantee their use and issues revolving around access to and prioritization of water use remain.

**Trachoma knowledge.** Results showed that there was increased knowledge around trachoma in villages receiving interventions in most EUs. Though in some cases it was minimal, the results imply that at minimum F&E programs were effective at increasing knowledge within communities, a factor in behavior change programs [22]. However, it is important to recognize that knowledge is one element. The study also looked at risk perception and noted there was little change in perception of risk across the different evaluation units. It is unclear if this lack of change in perception was because the respondents felt they were putting measures in place to reduce their risk and therefore they were not concerned or if they consistently did not feel trachoma was an issue in their communities.

**F&E related behaviors.** Though the percentage of children and adults with a clean face was at least 60% across the five community-based EUs, it was not expected that the post-survey results would show a significant decrease across all age groups and regions. This could be due to a number of reasons. For example, data collection was not always standardized in the months and time of day when data was collected. The months of the pre and post-surveys varied, with some data collection occurring during the rainy season and some in the dry season when water is typically less available. This could have impacted availability of water for hand and face washing use. The time of day when clean face was documented also varied as survey teams began at one house and moved throughout the village throughout the day. Other studies have shown that time of day and physical location of data collection may impact the likelihood of a face being clean [23–25].

Ultimately the survey results from schools and households showed that though there was an increase in knowledge at the school and household level, and, in some cases, an increase in presence of hand/face washing stations, this did not always result in a measurable change in trachoma prevention behaviors such as facial cleanliness at the household level. This shows that the interventions used were effective in increasing awareness of trachoma prevention, which is a first step to changing behavior, but there remains a gap either to translate that knowledge into changes in behavior or to measure the behaviors effectively. It may also show that the data collection itself needs improvement or that the period of intervention was not long enough to measure significant changes.

# **Challenges and successes**

The pre and post-surveys were powered to measure changes within the EU and did not measure the success or failure of specific F&E interventions used within EUs-most of which had multiple partners and heterogenous intervention design. This means we could not compare specific F&E interventions across districts, EUs, or countries to determine what did or did not work. It is also unknown if the F&E activities were truly implemented as intended, as the only point of reference were quarterly reports submitted by F&E implementing partners to the donors. Additionally, the surveys did not include trachoma infection data, therefore, these surveys cannot claim that particular F&E related activities directly led to a decrease in trachoma prevalence.

Throughout the period of implementation of F&E activities there was not a consensus on the percentage change needed within each indicator to determine if the programs were achieving success. For example, in Tanzania, the percentage of household respondents who knew one or more ways that trachoma spreads increased statistically significantly from 22.8% to 35.0%; however, programmatically 35% would be considered a sub-optimal achievement following almost two years of F&E related programming. On the other end of the spectrum, there were instances when the baseline was already high, such as facial cleanliness in schools or household levels of sanitation. Programmatically having over 90% of school children with a clean face, greater than 78% of households having access to latrines, and 88% of households free of human feces would be considered a high level of sanitation coverage in trachoma endemic regions. For indicators that began with a high baseline, it was difficult to power statistical evidence of increase [26]. The challenge therefore is determining what the minimal thresholds are for WASH related indicators rather than purely focusing on percentage change or statistically significant changes during evaluations.

There were multiple school related indicators in the FEME. These included children having a clean face; students having access to hand/face washing stations with water and soap present; schools having functional clean latrines for staff and students; improved water sources located on premises and accessible to all users during school hours; and an awareness about how to prevent and treat trachoma. These indicators are still viewed by the authors as adequate indicators to measure F&E school-based programming for trachoma prevention and education.

There were challenges implementing the pre and post-surveys within the schools. While observations were attempted in every school, observations could not be made in schools that had no water supply or no hand/face washing facilities. In this situation, observations were made for an hour and if no hand or face washing occurred the observer proceeded to help with the school questionnaires. There was no provision in the pre-survey data collection form for recording the absence of handwashing when a student used the latrine and did not wash their hands afterwards. The student was only recorded when they did wash their hands. Therefore, a comparison between handwashing and not handwashing could not be calculated. In the post-survey, the observation form was updated and allowed for individual observations to be collected in addition to collecting when a hand/face washing event should have occurred but did not (i.e. going to the toilet but not washing hands).

In enumeration of school WASH facilities, latrines/toilets were not assessed individually for each characteristic, rather, there was a count of all latrines meeting each characteristic. This means for percentage of schools with clean and functional latrines we could not look at both clean and functional latrines together and therefore could only pick one indicator. We chose to report on cleanliness of latrines only. This decision was made due to the survey design. There were many elements that made up a functional latrine and because we did not assess each latrine individually, we could not say, for example, if the one latrine counted for having a super structure was also the one latrine counted for having a drop cover. Therefore, we selected clean latrines because that was a stand-alone question included in the survey (i.e. how many of the latrines are clean). The 'percentage of children washing their faces when washing their hands during the school day' is not a critical data collection point as it does not inform if the face needed to be washed. Children not washing their face during this observed moment does not directly imply dirtier faces or an improper behavior. We would therefore not recommend this indicator but rather simply 'percentage of children with clean faces' at defined observation points, such as arrival at the school in the morning and before leaving school to return home. There were lessons learned from the data collection process during the pre-survey that were implemented for the post-survey to improve data quality and assurance. This included allowing data collectors and supervisors to check the data before sending to ensure increased data quality and control. In addition, mobile data capture forms were designed to validate eligibility criteria before allowing enumerators to proceed with data collection. Based on the experience from the pre-survey, a supervisor form was developed for the post-survey to calculate house-hold replacement rates and capture the population of the village for weightings (S7 Table).

In all three countries, F&E activities funded by the Trust and DFID began before the presurveys were conducted, due to funding dynamics and the time it took to get survey protocols developed, approved, and implemented. This delay in pre-survey implementation creates a limitation in that the surveys might not have detected some of the changes produced by the interventions in the months before the pre-survey was conducted.

Though the F&E programs were implemented for multiple years in each country, pre and post-surveys were conducted 16 months apart in Malawi, 13 months apart in Tanzania, and 20 months apart in Uganda. As behavior change is a long process, this is likely not enough time to measure a change in behavior. The surveys were expensive due to sample sizes across multiple regions and long household survey questionnaires that collected data on a range of topics, much of which was ultimately unused or was not processed in time to be of use to programming.

#### Recommendations

Based on the experience of selecting, implementing, monitoring, and evaluating F&E activities, there are several recommendations. First, the F&E activities chosen for implementation in these three countries were based on donor funding, implementing partner presence and perceptions of what was achievable in three years. Though these factors are important for program implementation, it is recommended that behavior change theory also be used to develop programs and M&E. This will not only help tailor programming to the specific needs of each community but also aid in the consistent and clear measurement directly linked to the activities. Second, ideally any future pre-surveys should occur before F&E activities are implemented if time and resources allow; however, in areas that have on-going F&E related activities, this might not be possible. Third, programs should determine what their output targets are before implementing F&E programs and when determining methods of measuring success. This should include not only change from pre and post-survey but also pre-identification of indicator levels which are considered optimal and which the maintenance of (as opposed to statistically significant increase) would be considered a successful outcome. Fourth, in order to evaluate whether activities are effective, it is important to know if they are being implemented as intended. Routine monitoring of activity implementation is critical and can provide insights and add depth to results from program evaluations. The FEME (S1, S2 and S3 Tables) provides a document that could be used as an example on which to improve. Fifth, there should be careful consideration of possible confounders of seasonality and time of day that data is collected, with efforts made to standardize the months and time of day data is collected. Sixth, unless multiple surveys are being conducted to measure longitudinal data, at least 24 months should pass between pre- and post-surveys allowing for more time for behavioral change practices to take root. Lastly, where possible, instead of conducting a standalone F&E survey, F&E data should be collected in connection with other planned programmatic, demographic, or health surveys in order to streamline human and financial resources.

# Conclusion

The application of the FEME in Malawi, Tanzania, and Uganda, and the implementation of a pre and post-survey to measure change for select F&E indicators represents an effort to fill a

gap in understanding how best to evaluate F&E activities in trachoma programs. It is clear from the lessons learned and recommendations that the FEME framework, survey indicators, and survey methodologies could use some improvement or modifications to make monitoring and evaluation of F&E activities more effective. Additionally, a more robust system for monitoring implementation of F&E activities would have aided in programs making quicker programmatic decisions and allowing a better understanding of the results of the post-surveys. Despite the limitations, the experience gained from implementing the FEME, pre and postsurveys, and the supplemental materials provided in this manuscript contribute towards the effort to progress our understanding of how best to evaluate F&E activities.

# **Supporting information**

S1 Text. Program Achievements. (DOCX)
S1 Table. Uganda FEME. (XLSX)
S2 Table. Malawi FEME. (XLSX)
S3 Table. Tanzania FEME. (XLS)
S4 Table. Household Survey.

(DOCX)

**S5 Table. School Survey.** (DOCX)

**S6 Table. Student Observation.** (DOCX)

**S7 Table. Supervisor Forms.** (XLSX)

# Acknowledgments

We would like to thank all the personnel who supported implementation of these surveys. This includes the community and school respondents, data collectors, village guides, technical and logistical support staff, and Cade Howard. Special thanks to the Malawi, Tanzania, and Uganda government ministries, and water and sanitation organizations who supported the F&E work: AMREF, Circle for Integrated Community Development (CICOD), Concern International, Heart to Heart, Helene Keller International, Johns Hopkins Center for Communication Programs, Matama, Simavi, Water Aid, Water Mission, and World Vision. We would also like to dedicate this article to the late Ruth Tushemereirwe who helped make the surveys in Uganda possible and worked tirelessly towards trachoma elimination in Uganda.

# **Author Contributions**

Conceptualization: Angelia M. Sanders, Ruth Dixon, Geordie Woods, Leah Wohlgemuth.

Data curation: Ruth Dixon, Logan Stuck, Geordie Woods.

Formal analysis: Ruth Dixon, Logan Stuck, Geordie Woods.

Funding acquisition: Angelia M. Sanders, Michaela Kelly.

- Investigation: Angelia M. Sanders, Ruth Dixon, Geordie Woods, Edridah M. Muheki, Gilbert Baayenda, Michael Masika, Holystone Kafanikhale, Upendo Mwingira, Leah Wohlgemuth.
- Methodology: Ruth Dixon, Geordie Woods, Leah Wohlgemuth.
- **Project administration:** Angelia M. Sanders, Michaela Kelly, Geordie Woods, Edridah M. Muheki, Gilbert Baayenda, Michael Masika, Holystone Kafanikhale, Upendo Mwingira, Leah Wohlgemuth.

Resources: Michaela Kelly.

- **Supervision:** Angelia M. Sanders, Michaela Kelly, Edridah M. Muheki, Gilbert Baayenda, Michael Masika, Holystone Kafanikhale, Upendo Mwingira, Leah Wohlgemuth.
- Validation: Angelia M. Sanders, Ruth Dixon, Geordie Woods, Leah Wohlgemuth.
- Visualization: Angelia M. Sanders, Ruth Dixon, Geordie Woods, Leah Wohlgemuth.
- Writing original draft: Angelia M. Sanders, Ruth Dixon, Logan Stuck, Michaela Kelly, Geordie Woods, Leah Wohlgemuth.
- Writing review & editing: Angelia M. Sanders, Ruth Dixon, Logan Stuck, Michaela Kelly, Geordie Woods, Edridah M. Muheki, Gilbert Baayenda, Michael Masika, Holystone Kafanikhale, Upendo Mwingira, Leah Wohlgemuth.

#### References

- 1. World Health Organization. Weekly Epidemiological Record. 2019; 94(29):317–328.
- 2. West S, Munoz B, Turner VM, Mmbaga BB, Taylor HR. The epidemiology of trachoma in central Tanzania. Int J Epidemiol. 1991; 20(4):1088–92. https://doi.org/10.1093/ije/20.4.1088 PMID: 1800408
- 3. Taylor HR, Velasco FM, Sommer A. The ecology of trachoma: an epidemiological study in southern Mexico. Bull World Health Organ. 1985; 63(3):559–67. PMID: <u>3876172</u>
- Sanders AM, Stewart AEP, Makoy S, Chebet JJ, Magok P, Kuol A, et al. Burden of trachoma in five counties of Eastern Equatoria state, South Sudan: Results from population-based surveys. PLoS Neglected Tropical Diseases. 2017;(6). https://doi.org/10.1371/journal.pntd.0005658 PMID: 28614375
- Munoz B, West S. Trachoma: the forgotten cause of blindness. Epidemiol Rev. 1997; 19(2):205–17. https://doi.org/10.1093/oxfordjournals.epirev.a017953 PMID: 9494783
- Emerson P, Cairncross S, Bailey RL, Mabey DC. Review of the evidence base for the 'F' and 'E' components of the SAFE strategy of trachoma control. Trop Med Int Health. 2000; 5:515–27. <a href="https://doi.org/10.1046/j.1365-3156.2000.00603.x">https://doi.org/10.1046/j.1365-3156.2000.00603.x</a> PMID: 10995092
- Emerson P, Bailey RL, Walraven GE, Lindsay SW. Human and other faeces as breeding media of the trachoma vector Musca sorbens. 2001; 15(3):314–20.
- Robinson A, Bristow J, Holl MV, Makalo P, Alemayehu W, Bailey RL, et al. et al. Responses of the putative trachoma vector, *Musca sorbens*, to volatile semiochemicals from human faeces. PLoS Neglected Tropical Diseases. 2020; 14(3): e0007719. <u>https://doi.org/10.1371/journal.pntd.0007719</u> PMID: 32126087
- Cromwell E, Emerson P, Courtright P. Women and Trachoma: Achieving Gender Equity in the Implementation of SAFE. February 2009. The Carter Center, Kilimanjaro Centre for Community Ophthalmology, The Elfenworks Foundation. Available from: https://www.cartercenter.org/resources/pdfs/health/ trachoma/women\_trachoma.pdf
- Abdala M, Singano CC, Willis R, Macleod CK, Backers S, Flueckiger RM, et. al. The Epidemiology of Trachoma in Mozambique: Results of 96 Population-Based Prevalence Surveys. Ophthalmic epidemiology. 2018; 25(sup1): 201–210. https://doi.org/10.1080/09286586.2017.1351996 PMID: 28910562
- Greenland K, White S, Sommers K, Biran A, Burton MJ, Sarah V, et al. Selecting behaviour change priorities for trachoma 'F' and 'E' interventions: A formative research study in Oromia, Ethiopia. PLoS Neglected Tropical Diseases. 2019; 13(10):1–19.

- Lynch M, West S, Munoz B, Kayongoya A, Taylor HR, Mmbaga BB. Testing a participatory strategy to change hygiene behaviour: face washing in central Tanzania. Trans R Soc Trop Med Hyg. 1994; 88(5) 513–7. https://doi.org/10.1016/0035-9203(94)90141-4 PMID: 7992324
- Queen Elizabeth Diamond Jubilee Trust. A Lasting Legacy. 2020 [cited January 9 2020]. Available from: https://www.jubileetribute.org/assets/images/A-Lasting-Legacy.pdf
- 14. Sightsavers. Experts celebrate progress towards eliminating trachoma in Tanzania. 2019 [cited September 23 2021]. Available from: https://www.sightsavers.org/news/2019/02/the-trust-trachoma-tanzania/
- 15. Johns Hopkins University. Protocol and Methods for Trachoma Situation Analysis. Johns Hopkins Bloomberg School of Public Health Center for Communication Programs. 2014. Available from: https:// www.trachomacoalition.org/sites/default/files/content/resources/files/Situation%20Analysis% 20Protocol.pdf
- 16. International Coalition for Trachoma Control. All you need for F&E: A practical guide to partnering and planning. International Coalition for Trachoma Control. 2015. Available from: https://www.trachomacoalition.org/sites/default/files/content/resources/files/All%20you%20need%20for%20FandE%20-%20a%20toolkit%20for%20planning%20and%20partnering\_web\_incl\_tools\_0.pdf
- Waite RC, Woods G, Velleman Y, Freeman MC. Collaborating to develop joint water, sanitation and hygiene (WASH) and neglected tropical disease (NTD) sector monitoring: an expert consultation. International Health. 2017; 9(4):215–225. https://doi.org/10.1093/inthealth/ihx008 PMID: 28407112
- West SK, Ansah D, Munoz B, Funga N, Mkocha H. The "F" in SAFE: Reliability of assessing clean faces for trachoma control in the field. PLoS Neglected Tropical Disease. 2017; 11(11). <u>https://doi.org/ 10.1371/journal.pntd.0006019</u> PMID: 29190298
- 19. JMP. Hygiene. 2021. Available from: https://washdata.org/monitoring/hygiene.
- 20. LINKS, Task Force for Global Health, Atlanta, GA, USA; https://linkssystem.org/24
- 21. CommCare, Dimagi, Cambridge, MA, USA; https://www.dimagi.com/commcare
- 22. Ram P. Practical Guidance for Measuring Handwashing Behavior: 2013 Update. World Bank: Water and Sanitation Program. 2013. Available from: https://www.wsp.org/sites/wsp/files/publications/WSP-Practical-Guidance-Measuring-Handwashing-Behavior-2013-Update.pdf
- 23. Harding-Esch EM, Holland MJ, Schemann J, Sissoko M, Sarr B, Butcher R, et al. Facial cleanliness indicators by time of day: results of a cross-sectional trachoma prevalence survey in Senegal. Parasites & Vectors. 2020; 13(1):556. https://doi.org/10.1186/s13071-020-04410-w PMID: 33203456
- Zack R, Mkocha H, Zack E, Munoz B, West S. Issues in defining and measuring facial cleanliness for national trachoma control programs. Trans R Soc Trop Med Hyg. 2008; 102(5):426–31. <u>https://doi.org/ 10.1016/j.trstmh.2008.02.001</u> PMID: 18346769
- King JD, Ngondi J, Kasten J, Diallo MO, Zhu H, Cromwell EA, et al. Randomised trial of face-washing to develop a standard definition of a clean face for monitoring trachoma control programmes. Trans R Soc Trop Med Hyg. 2011; 105(1):7–16. https://doi.org/10.1016/j.trstmh.2010.09.008 PMID: 21036378
- 26. Garn JV, Sclar GD, Freeman MC, Penakalapati G, Alexander KT, Brooks P, et al. The impact of sanitation interventions on latrine coverage and latrine use: A systematic review and meta-analysis. International Journal of Hygiene and Environmental Health. 2017; 220(2 Pt B): 329–340. https://doi.org/10. 1016/j.ijheh.2016.10.001 PMID: 27825597